Authors reported the title: FAST-TRACK PROTOCOLS IN LAPAROSCOPIC LIVER SURGERY. APPLICABILITY AND CORRELATION WITH DIFFICULTY SCORING SYSTEMS. It is clinically significant and new to the extent of some degree. And some questions came forth.

1. The manuscript should be concise and much syntax should be polished.
R: Thanks for the comment. We have extensively reduced the length of the manuscript and checked all syntax.

2. The data of operated patients were missing, major or minor resections were invisible, so conclusion is not evidenced.
R: Thanks for the comment. We have included all baseline data on Table 1 as the reviewer suggested. It has clearly improved the quality of the manuscript.